Author's response to reviews

Title: WalkMore: A randomized controlled trial of pedometer-based interventions differing on intensity messages

Authors:

Catrine Tudor-Locke (Tudor-Locke@pbrc.edu)
Damon L Swift (swiftd@ecu.edu)
John M Schuna Jr. (John.Schuna@pbrc.edu)
Amber T Dragg (Amber.Dragg@pbrc.edu)
Allison B Davis (Allison.Davis@pbrc.edu)
Corby K Martin (Corby.Martin@pbrc.edu)
William D Johnson (William.Johnson@pbrc.edu)
Timothy S Church (Timothy.Church@pbrc.edu)

Version: 2
Date: 27 January 2014

Author's response to reviews:

January 27, 2014

Please note the itemized responses to your email dated Jan 22, 2014:

1. We sent an email to BMCSeriesEditorial@biomedcentral.com with attached Ethical and Funding Approval Documentations.
2. We say "This work was supported by an award from the American Heart Association" in the competing interests section of our manuscript. As part of the AHA award process, the study protocol underwent peer-review by this funding body
3. The study is currently still recruiting and collecting data.
4. No publications have been published or submitted to any journal.
5. We have added TRN to the end of our abstract in the submission system.
6. We confirm that the email addresses of all the authors are now in the title page.